Author's response to reviews

Title: Total phenolic and flavonoid content and antibacterial activity of Punica granatum L. var. pleniflora flowers (Golnar) against bacterial strains causing foodborne diseases

Authors:

Arash Mahboubi (a.mahboubi@sbmu.ac.ir)
Jinous Asgarpanah (taxolfa@yahoo.com)
Parisa Nosrati Sadaghiyani (parisa.nosrati8986@gmail.com)
Mehrdad Faizi (m.faizi@sbmu.ac.ir)

Version: 6 Date: 17 September 2015

Author's response to reviews:

Dear editor,

We would like to thank you for your kind consideration and the valuable comments and suggestions. We believe your comments will help us to improve our manuscript. We have considered each of the reviewer's comments and provide our answers below. We hope that this improved manuscript is acceptable for publication in your valuable journal.

1. As the reviewer requested in Table 5 the comparisons letters over all the values for each column were added.

Thank you in advance for your kind consideration and timely review of this manuscript.

Sincerely Yours,

Arash Mahboubi, Pharm. D., Ph. D.

Department of Pharmaceutics

School of Pharmacy

Shahid Beheshti University of Medical Sciences

No. 2660, Vali-e-Asr Ave., Tehran 1996835113, Iran

P.O. Box: 14155-6153

Tel/Fax: +98 21 88 200068

Email: a.mahboubi@sbmu.ac.ir